Author's response to reviews

Title: A pilot study of an individualized comprehensive treatment for advanced gastric cancer with para-aortic lymph node metastases

Authors:

Qi He (heqinju@gmail.com)
Long Ma (1005288306@qq.com)
Yang Li (li_yang82@icloud.com)
Guoli Li (liguoli15@163.com)

Version: 8
Date: 12 December 2015

Author's response to reviews: see over
A pilot study of an individualized comprehensive treatment for advanced gastric cancer with para-aortic lymph node metastasis

Dear editors,
The name of the IRB is the Independent Institutional Review Board (IRB) of Jinling Hospital. We state it in ethical approval.
Thank you very much.

Sincerely
Qi He
On behalf of Dr. Guoli Li